**Author’s response to reviews**

**Title:** Evaluation of serum prolidase activity and oxidative stress markers in men with BPH and prostate cancer

**Authors:**

Faruk Kucukdurmaz (farukdr@hotmail.com)
erkan efe (drerkanefe@hotmail.com)
Ahmet Celik (drahmetcelik@gmail.com)
hasan dagli (hasandagli63@gmail.com)
metin kilinc (mkilinc@ksu.edu.tr)
sefa resim (sefaresim@gmail.com)

**Version:** 2 Date: 19 Oct 2017

**Author’s response to reviews:**

Dear reviewer.

Thank you for your valuable contributions. As you requested, the presentation of the bar graph was rearranged. Thank you again for having your merit time to evaluate our article.